Author's response to reviews

Title: Gastric transit and small intestinal transit time and motility assessed by a magnet tracking system

Authors:

Jonas Worsoe (jonas.worsoe@gmail.com)
Lotte Fynne (lfynne@hotmail.com)
Tine Gregersen (tine_gregersen@hotmail.com)
Vincent Schlageter (vincent.schlageter@motilis.com)
Lisbet A. Christensen (lisbchte@rm.dk)
Jens F. Dahlerup (jensdahl@rm.dk)
Nico J.M. Rijkhoff (nr@hst.aau.dk)
Søren Laurberg (SOERLAUR@rm.dk)
Klaus Krogh (klaukrog@rm.dk)

Version: 8 Date: 29 December 2011

Author's response to reviews:

To the Production team
Now I think I finally managed
Happy New Year
Kind Regards Jonas Worsoe